**Author's response to reviews**

**Title:** Clinical outcomes of seasonal influenza and pandemic influenza A (H1N1) in pediatric inpatients

**Authors:**

Pranita D Tamma (ptamma1@jhmi.edu)
Alison Turnbull (aturnbul@jhsph.edu)
Aaron M Milstone (amilsto1@jhmi.edu)
Sara E Cosgrove (scosgro1@jhmi.edu)
Alexandra Valsamakis (avalsam1@jhmi.edu)
Alicia Budd (abudd3@jhmi.edu)
Trish M Perl (tperl@jhmi.edu)

**Version:** 5  **Date:** 21 September 2010

**Author's response to reviews:** see over